Author’s response to reviews

Title: Fulminant herpes simplex hepatitis secondary to tongue piercing in an immunocompetent adult: a case report

Authors:

Shaheen E Lakhan (slakhan@gnif.org)
Lindsey Harle (lharle@gnif.org)

Version: 5 Date: 30 June 2008

Author’s response to reviews:

To the Editors:

We have revised our manuscript and addressed all language issues, re-organized and shortened the discussion/conclusion, and incorporated suggested references. Also, we specifically included the HSV type indentified and the timing of the patient’s piercing. Thank you.

Sincerely,
Shaheen Lakhan and Lindsey Harle